Author’s response to reviews

Title: Prevalence and key radiographic spinal malalignment parameters that influence the risk for gastroesophageal reflux disease in patients treated surgically for adult spinal deformity

Authors:

Tetsuro Ohba (tooba@yamanashi.ac.jp)
Shigeto Ebata (sebata@yamanashi.ac.jp)
Kensuke Koyama (kkoyama@yamanashi.ac.jp)
Hirotaka Haro (haro@yamanashi.ac.jp)

Version: 1 Date: 06 Oct 2017

Author’s response to reviews:

Thank you very much for the excellent reviews. We hereby resubmit this document with point-by-point responses to all of the editor’s comments.

Technical Comments:

1. Please upload a clean revised manuscript that does not contain any tracked changes or highlighting as your main article file.
   →We fixed this point.

Editor’s comments:

1. Please amend your abstract to ensure that it is less than 350 words in total.
   →We fixed this point.

2. Please amend your sub-heading ‘Introduction’ to ‘Background’ within your manuscript.
   →We fixed this point.
3. Please confirm in your ethics statement whether the participant consent was written or verbal. If it was verbal, please confirm that you received ethics committee approval for verbal consent and detail how you documented the consent.

4. Please add your complete ethics statement to the relevant section under the Declarations subheading.

   → This study was approved by our institutional review board by ethical committee of University of Yamanashi School of Medicine (Application number 1183). Additionally, we received informed consent from all eligible patients. Now these information have been provided in the section on Ethics approval.